Author's response to reviews

Title: A simulation study of sample size for multilevel logistic regression models

Authors:

Rahim Moineddin (rahim.moineddin@utoronto.ca)
Flora I Matheson (mathesonf@smh.toronto.on.ca)
Richard H Glazier (rick.glazier@ices.on.ca)

Version: 6 Date: 13 July 2007

Author's response to reviews:

Thank you for your assistance with formatting our paper.
We made the following changes:

Sentence cases are used for title.
The text "Table about here" is removed.
References are amended.
Tracking are removed.
"Reference List" is changed to "references".
The heading "Tables" are inserted prior to table section.

Regards,

Rahim Moineddin